Author's response to reviews

Title: Mechanisms of ring chromosome formation, ring instability and clinical consequences.

Authors:

Roberta S Guilherme (robertaguilherme@hotmail.com)
Vera F Ayres Meloni (vera.meloni@gmail.com)
Chong A Kim (chong.kim@icr.usp.br)
Renata Pellegrino (renata.pellegrino@gmail.com)
Sylvia S Takeno (sylvinha66@yahoo.com.br)
Nancy B Spinner (spinner@mail.med.upenn.edu)
Laura K Conlin (conlinl@email.chop.edu)
Denise M Christofolini (denise.morf@gmail.com)
Leslie D Kulikowski (lesliekulik@gmail.com)
Maria I Melaragno (melaragno.morf@epm.br)

Version: 6 Date: 16 December 2011

Author's response to reviews: see over
Subject: Revised Manuscript: Mechanisms of ring chromosome formation, ring instability and clinical consequences

December 16, 2011

Dear

We were glad to know our MS “Mechanisms of ring chromosome formation, ring instability and clinical consequences” has been accepted for publication in the BMC Medical Genetics journal.

Enclosed please find the revised version of the manuscript with major revisions.

- The names of the authors in the manuscript were changed and they are written exactly in the submission system.
- We want that the tables be visible within the final published manuscript so we included a table section following the references.
- We removed the first page of the manuscript.
- We changed the title “Materials and Methods” to “Methods”.
- We rearranged the figures according the journal style.

We hope that our paper now meets all the requirements for publication.

Thank you for all your attention.

Yours sincerely,

Roberta S Guilherme